Author's response to reviews

Title: Space-time clusters of breast cancer using residential histories: A Danish case-control study

Authors:

  Rikke Baastrup Nordsborg (baastrup@cancer.dk)
  Jaymie R Meliker (jrmeliker@gmail.com)
  Annette Kjær Ersbøll (ake@si-folkesundhed.dk)
  Geoffrey M Jacquez (gjacquez@buffalo.edu)
  Aslak H Poulsen (aslak@cancer.dk)
  Ole Raaschou-Nielsen (ole@cancer.dk)

Version: 3  Date: 3 April 2014

Author's response to reviews:

To the BMC Cancer Editorial Team,

Regarding the manuscript MS: 1710394923111980 - Space-time clusters of breast cancer using residential histories: A Danish case-control study

The following corrections have been made in the manuscript:

'BMJ Cancer', in the competing interest section has been changed to 'BMC Cancer'.

Regarding the origin of the maps in figures 1-4:

The authors produced the maps, but the Danish Geodata Agency and OpenStreetMap hold the copyrights for the original basemap data that were used to produce the maps. Figures can be reproduced, as long as the following sentences appear on the maps:

For Figure 1:
“Contains data from the Danish Geodata Agency.”

For Figures 2, 3 and 4:
“Contains data from the Danish Geodata Agency and © OpenStreetMap (and) contributors, CC-BY-SA.”

In the manuscript I have added these sentences to the legends of each of the 4 figures. I hope this is satisfactory.

Kind regards,

Rikke Baastrup Nordsborg